ABSTRACT

Shareholder engagement has come to be seen to be pivotal to good corporate governance. It is therefore more important than ever for the mechanisms of shareholder engagement to be up to the important task they are meant to perform. Historically, these mechanisms have fallen short because of the capital markets' necessary dependency on custody chains. Custody chains perform valuable functions but they also create a distance between the company and the end investor from which flows a significant risk of voting preferences and other important information not passing smoothly up and down the chain. Corporate governance is thus at risk of being distorted by process deficiencies.

Changes to the operation of custody changes that were introduced by the EU's flagship amended Shareholder Rights Directive II became operative in September 2020. This article is therefore among the first to draw on operational impact to assess the significance of these new measures. Experience has already demonstrated that while there have been some important advances, the regulatory changes are struggling on two levels: they have created new uncertainties within national laws; and they have failed to provide the necessary cross-border harmonization of key concepts. These are not issues that can be resolved effectively by technology or market workarounds on their own. There is a continuing need for legislative change to ensure that legal uncertainties do not act as a barrier to further technology-driven market standardization and innovation.

Keywords: corporate governance; engagement; shareholder rights; custody chains

JEL Codes: G34, G38, K22

Wordcount: 14468 (incl footnotes)

SHAREHOLDER ENGAGEMENT AND CUSTODY CHAINS*

1. INTRODUCTION

Custody chains complicate shareholder engagement by interposing several layers of intermediaries between companies and their end investors. Corporate governance-related problems with custody chains exist in major markets around the world. This article focuses on the situation in Europe, by which is meant the EU/EEA and also the UK.

The EU's flagship amended Shareholder Rights Directive II (2017) addresses the operation of custody chains. The changes became operative in September 2020 and this article is therefore among the first to draw on operational impact to assess the significance of the new measures. Unfortunately, experience has already demonstrated that while there have been some important advances, the regulatory changes are struggling on two levels: they have created new uncertainties within national laws; and they have failed to provide the necessary cross-border harmonization of key concepts. These are not issues that can be resolved effectively by technology or market workarounds on their own. Rather, they need to be addressed at the legislative level to ensure that legal uncertainties do not act as a barrier to further technology-driven market standardization and innovation.

The article proceeds as follows. Part 2 outlines the rise of shareholder engagement because this trend has made it all the more important to ensure that the mechanisms of shareholder engagement work well. Part 3 considers the importance of custody chains to the functioning of international capital markets and why, notwithstanding their many benefits, they can act as a barrier to effective shareholder engagement. Part 4 examines the new EU measures and the main problems that have emerged. Part 5 explains why we cannot look to technology to provide quick fixes to what are fundamental conceptual problems and identifies key areas that require legislative intervention. Part 6 concludes.

2. THE RISE OF SHAREHOLDER ENGAGEMENT

Shareholder engagement has become an increasingly important part of corporate governance since the start of the 21st century. This trend towards greater emphasis on shareholder engagement is attributable to a mix of business and regulatory factors. The following sections elaborate.

Why activist, active and passive funds may choose to engage

Engagement with investee companies to increase their value emerged by the mid-2000s as a core part of the business strategy of activist hedge fund investors. Engagement has always

^{• ...}I am grateful to Pierre- Henri Conac, James Cunningham and Edwin De Pauw for their insightful comments and suggestions. All errors that remain are my own.

¹ Brian R Cheffins and John Armour, 'The Past, Present, and Future of Shareholder Activism by Hedge Funds' (2011) 37 Journal of Corporation Law 51; Anna L Christie, 'The New Hedge Fund Activism: Activist Directors and the Market for Corporate Quasi-Control' (2019) 19 Journal of Corporate Law Studies 1; Marco Becht and others, 'Returns to Hedge Fund Activism: An International Study' (2017) 30 The Review of Financial Studies

been key to the investment strategies of private equity and venture capital funds.² General investment funds are also looking at engagement as a way both to distinguish themselves from competitors and to improve investment returns.³ Active fund managers whose business model involves actively buying, selling and holding securities are investing in building up engagement capabilities (in Hirschman's classic framework, using 'voice' to complement 'exit').5

Active funds are being challenged competitively by the rise of lower-cost index funds whose portfolios are constructed to track the components of a financial market index. 6 (The market share of index funds in the EU stands at just 16 per cent compared to over 50 per cent for equity-linked US mutual funds but enhanced European growth in passive vehicles is anticipated, with predictions of a 30% share during this decide, of which a significant proportion is expected to be in exchange traded funds.⁷) Superficially, the rise of passive tracker funds could be viewed as a trend that runs counter to the growing emphasis on engagement because the low-cost business model suggests a strong likelihood of underinvestment in stewardship. On the other hand, while the low cost business model of passive funds militates against investment in engagement, being tied to the index removes the option of discretionary exit and thus suggests a stronger dependency on engagement/voice than active funds which may limit their investment in engagement precisely because they have the option of exit available to them.

An obvious disincentive for any fund – active or passive – to engage in engagement activities is that it will not fully capture the benefit of that activity to the exclusion of other 'free rider' investors. There is a growing body of empirical literature exploring whether there is under-investment in stewardship by funds due to weak incentives and costs.⁸ Notwithstanding the widespread scepticism about incentives to invest, there is some evidence that funds do view stewardship capabilities as a source of competitive advantage. 9 With growing public understanding of the climate crisis changing what individuals expect of the institutional investors that manage their long-term savings, managers of funds are likely to see more reason to burnish their credentials as agents for positive change on ESG through

2933; Alessio M Pacces, 'Hedge Fund Activism and the Revision of the Shareholder Rights Directive' (2017) European Corporate Governance Law Working Paper No 353/2017

https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2953992>.

² FCA, Building a Regulatory Framework for Effective Stewardship DP19/1 (2019) para 5.16.

³ Elroy Dimson, Oğuzhan Karakaş and Xi Li, 'Active Ownership' (2015) 28 The Review of Financial Studies

⁴ Albert O Hirschman, Exit, Voice, and Loyalty (Harvard University Press 1970).

⁵ Marco Becht, Julian Franks and Hannes F Wagner, 'Corporate Governance Through Voice and Exit' (2019)

⁶ On the differences between active and index (passive) fund models: Lucian A Bebchuk, Alma Cohen and Scott Hirst, 'The Agency Problems of Institutional Investors' (2017) 31 Journal of Economic Perspectives 89, 94-95.

⁷ Sylvain Broyer and David Henry Doyle, 'The EU Capital Markets Union: Turning The Tide'

https://www.spglobal.com/en/research-insights/featured/the-eu-capital-markets-union-turning-the-tide.

⁸ Including Bebchuk, Cohen and Hirst (n 6); Lucian Bebchuk and Scott Hirst, 'Index Funds and the Future of Corporate Governance: Theory, Evidence and Policy' (2019) 119 Columbia Law Review 2029; Ian R Appel, Todd A Gormley and Donald B Keim, 'Passive Investors, Not Passive Owners' (2016) 121 Journal of Financial Economics 111; Holger Spamann, 'Indirect Investor Protection: The Investment Ecosystem and Its Legal Underpinnings' forthcoming (ECGI working paper); Becht, Franks and Wagner (n 5); Bobby Reddy, 'The Emperor's New Code? Time to Re-Evaluate the Nature of Stewardship Engagement Under the UK's Stewardship Code' (2021) 84 Modern Law Review 842.

⁹ FCA (n 2), para 5.5.

shareholder engagement. ¹⁰ It makes sense for funds with diversified portfolios to direct their stewardship efforts towards market-wide externalities. ¹¹

Retail investors wanting a 'say'

After years of decline, ¹² direct retail investor in equities and participation in voting and other forms of shareholder engagement is also experiencing an upturn in some countries, ¹³ although this has not yet extended to the EU. ¹⁴ Overall, retail investment represents a small portion of the capital markets and retail participation in voting is at a much lower level than that of institutional and professional investors ¹⁵ but these considerations notwithstanding, recent study focused on the US market has found that retail shareholders can be an influential voting bloc. ¹⁶ The European Commission has recently cited growing interest among citizens, especially young people, in having a say in how companies are run, notably as regards sustainability issues, as a factor shaping its policy thinking on the operation of custody chains. ¹⁷ A campaign to give shareholders a 'say on climate' akin to their 'say on pay' is attracting some powerful advocates including the UN Secretary-General and a former Governor of the Bank of England, and some prominent companies such as Unilever, Glencore and Moody's are already doing this. ¹⁸ The stunning success in 2021 of Engine No 1, a small activist investment fund, in winning a proxy contest for three board seats at Exxon

 $\underline{https://www.ons.gov.uk/economy/investmentspensions and trusts/bulletins/ownership of ukquoted shares/2018}$

¹⁰ Portfolio International, 'EFTs and ESG Engagement: Passively Active' (November 2020) (https://www.portfolio-institutional.co.uk/esg-hub/etfs-and-esg-engagement-passively-active/, quoting, among others, Marta Jankovic, a director in BlackRock's Investment Stewardship team as saying "It is a misconception that index managers do not have the tools to drive change because they cannot sell stocks that may not fulfil their sustainability expectations ... "Through engagement and voting we express our views as shareholders and help drive long-term change within companies," she adds. Through engagement we build an understanding of a company's approach to governance and its sustainable business practices. We also communicate our views publicly, so companies know what we expect of them."

¹¹ Dimson, Karakaş and Li (n 3); Madison Condon, 'Externalities and the Common Owner' (2020) 95 Washington Law Review 1; Jeffrey N Gordon, 'Systematic Stewardship' (2021) European Corporate Governance Institute Law Working Paper No 566/2021

https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3782814; Luca Enriques and Alessandro Romano, 'Rewiring Corporate Law for an Interconnected World' (2022) 64 Arizona Law Review forthcoming.

¹² Kobi Kastiel and Nili Yaron, 'In Search of the "Absent" Shareholders: A New Solution to Retail Investors' Apathy' (2016) 41 Delaware Journal of Corporate Law 55.

¹³ The most recent publicly available data for the UK (2018) show the proportions of UK shares beneficially owned by UK-resident individuals rose to 13.5%, up by 1.2 percentage points from 2016, moving further away from the historical low of 10.2% in 2008: Office for National Statistics 'Ownership of UK Quoted Shares' (14 January 2020)

¹⁴ Better Finance, a body that advocates for the interests of European savers and investors, argues that the decline over decades in the number of individual shareholders and their share of EU listed companies is linked to EU policies not being favourable overall to the development of individual share ownership. EU policy emphasis remains mostly on promoting retail participation in the capital markets via pooled products: High Level Forum on Capital Markets Union, 'A New Vision for Europe's Capital Markets - Final Report' https://ec.europa.eu/info/sites/default/files/business_economy_euro/growth_and_investment/documents/200610-cmu-high-level-forum-final-report_en.pdf, 18-21; European Commission, 'A Retail Investment Strategy for the EU – Roadmap' https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/12755-EU-strategy-for-retail-investors_en.

¹⁵ 32% compared to 80% according to a recent study of US retail shareholder voting behaviour: Alon Brav, Matthew D Cain and Jonathan Zytnick, 'Retail Shareholder Participation in the Proxy Process: Monitoring, Engagement, and Voting' forthcoming.
¹⁶ ibid.

¹⁷ European Commission, 'A Capital Markets Union for People and Businesses-New Action Plan' COM(2020) 590, 13.

¹⁸ Taken from https://www.sayonclimate.org/

Mobil via a relatively modest \$30 million campaign that included a website, Twitter posts, employee forums and television appearances, was a high-profile example of how coordinated efforts by investors can alter the corporate governance landscape. Over time, the development of blockchain-based and/or application programming interface (API) digital technologies could help retail investors to become an increasingly important part of this trend. On the control of the co

Shareholder engagement as an instrument of financial regulatory policy

On the regulatory side, investment intermediaries' duties to act in the interests of their clients and related supervisory expectations on what is needed for markets to function well increasingly encompass engagement with investee companies and the making of considered decisions about the exercise of voting discretions.²¹ Transparency obligations on institutional investors, asset managers and proxy advisers with respect to their engagement with investee companies, including mandatory disclosures relating to policies on voting and information about the actual casting of votes, are becoming progressively more demanding. The 2017 version of the Shareholder Rights Directive represented a step-change for Europe in this respect.²² (Changes to the operation of custody chains that were also introduced by the revising directive are the primary focus of this article.) New mandatory sustainability-related disclosure requirements for the investment industry are being introduced.²³ Baseline formal obligations, the details of which vary from jurisdiction to jurisdiction but which typically do not extend to a specific legal obligation to undertake stewardship activities (though general duties to act in the interests of clients may require consideration of stewardship in particular circumstances),²⁴ are bolstered by soft law 'comply or explain', 'apply and explain'

. .

Alexander I Platt, 'Beyond "Market Transparency": Investor Disclosure and Corporate Governance' (2022)
 Stanford Law Review forthcoming. More generally on the impact of coordinated engagement by large investors: Elroy Dimson, Oğuzhan Karakaş and Xi Li, 'Coordinated Engagements' (2021).

²⁰ Alexander Daniels, 'Blockchain & Shareholder Voting: A Hard Fork for 21st-Century Corporate Governance' (2018) 21 University of Pennsylvania Journal of Business Law 405, 439 (disccusing the future emergence of decentralized autonomous organizations for shareholder voting).

²¹ Scott Hirst and Adriana Z Robertson, 'Hidden Agendas in Shareholder Voting' forthcoming (US focused); Law Commission, 'Fiduciary Duties of Investment Intermediaries' Law Com No 350 (2014) (UK focused). The UK FCA links its interest in effective stewardship by investment intermediaries to its strategic objective to make markets function well, and sees stewardship activities as improving market quality and integrity, helping to create sustainable, long-term value for clients and beneficiaries, and as having wider economic, environmental and societal benefits: FCA (n 2) paras 3.1-3.11; FCA, Building a Regulatory Framework for Effective Stewardship – Feedback to DP19/1 FS19/7 (2019) para 1.1.

²² Directive 2017/828 of the European Parliament and of the Council of 17 May 2017 amending Directive 2007/36 as regards the encouragement of long-term shareholder engagement [2017] OJ L 132/1, arts 3g–3k. This article uses SRD II to refer to the consolidated revised version of the directive.

²³ Regulation (EU) 2020/852 on the establishment of a framework to facilitate sustainable investment [2020] OJ L198/13 (EU Taxonomy Regulation); Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on sustainability-related disclosures in the financial services sector [2019] OJ L317/1 (Sustainable Finance Disclosure Regulation (SFDR)). The EU Taxonomy Regulation and SFDR were not operative in the UK at the end of the Brexit transition period and were therefore not required to be onshored. The UK is developing its own green taxonomy and disclosure requirements based on adoption of the FSB's Tasforce for Climate-related Financial Disclosures (TCFD).

²⁴ Freshfields Bruckhaus Deringer, Principles for Responsible Investment, Generation Foundation and UNEP Finance Initiative, 'A Legal Framework for Impact' (2021) https://www.unpri.org/download?ac=13902 (major 11 jurisdiction study including detailed examination of the extent to which the law requires or permits investing for sustainability impact).

In 2021 the EU introduced a package of measures requiring different types of investment fund manager to integrate sustainability risks into the factors to be taken into account as part of their duties towards investors

stewardship codes that were first introduced to address perceived problems of short-termism and excessive passivity on the part of institutional investors.²⁵ The first stewardship code was published in the UK in 2010 and other countries, including the United States in 2017, have since adopted their own versions.²⁶ Transnational bodies have also produced stewardship codes.²⁷ Stewardship codes seek to set high standards for the investment intermediary ecosystem, by which is meant those that invest money on behalf of savers and pensioners and those that support them. The ecosystem includes pension fund trustees and insurers (in this context often described as asset owners), asset managers and service providers, such as proxy advisers and investment consultants. Doubts as to the effectiveness of codes in driving responsible stewardship and about their suitability for markets where the US/UK institutional investor-dominated share ownership model is not prevalent have not inhibited the global rise of stewardship codes.²⁸

3. CUSTODY CHAINS COMPLICATE THE ENGAGEMENT LANDSCAPE

Custody chains are largely a byproduct of the organization of modern system for trading, clearing and settlement of securities. Publicly-traded securities increasingly exist only in dematerialized, book entry, form and are typically held within central securities depositories (CSDs). Settlement of trades via a CSD results in securities being recorded to the account of a CSD member. Some CSDs allow individual investors to have their own direct account with the CSD but this a relatively unusual (and often quite expensive) option.²⁹ More typically, there will be a custody chain involving the CSD, the CSD member, such as a custodian bank, to whose account the securities are credited and then further intermediaries until the chain eventually reaches the end investor. Intermediaries typically use 'omnibus' accounts to hold securities on behalf of multiple investors.

These well-established market arrangements provide the cross-border links between national/regional financial market infrastructures that enable investors to hold internationally-diversified portfolios securely, efficiently and cost-effectively. For a typical investor, directly accessing a large number of national CSDs in order to access securities in different jurisdictions is unlikely to be a realistic option.³⁰ A further advantage of holding securities in

⁽details on these amending delegated acts on sustainability preferences, fiduciary duties and product governance are at https://ec.europa.eu/info/publications/210421-sustainable-finance-communication_en). This package stopped short of explaining whether investors had duties to consider undertake stewardship activities but in a new strategy document published a few month later, the European Commission acknowledged the need to do more to clarify the fiduciary duties and stewardship rules of investors to reflect the financial sector's contribution to Green Deal targets: European Commission, 'Strategy for Financing the Transition to a Sustainable Economy' COM(2021) 390.

²⁵ Dionysia Katelouzou and Dan W Puchniak, 'Global Shareholder Stewardship: Complexities, Challenges, and Possibilities' in Dionysia Katelouzou and Dan W Puchniak (eds), *Global Shareholder Stewardship* (Cambridge University Press forthcoming).

²⁶ The European Corporate Governance Institute maintains a database: https://ecgi.global/content/codes-stewardship.

²⁷ Katelouzou and Puchniak (n 25).

²⁸ ibid.

²⁹ eg, Euroclear Finland, Euroclear Sweden and Euroclear UK & International (which operates the CREST system) offer this option: https://www.euroclear.com/services/en/private-investor-services.html.

³⁰ Christopher Twemlow, 'Why Are Securities Held in Intermediated Form?' in Louise Gullifer and Jennifer Payne (eds), *Intermediation and Beyond* (Hart Publishing 2019).

CSDs via custodians is that this can provide access to intraday credit facilities and to a range of asset servicing and financial collateral intermediary services. However, despite their undoubted benefits, custody chains complicate shareholder participation in corporate governance by interposing several layers of intermediaries between companies and their end investors. The complexities are usually intensified in the case of cross-border chains, both because such chains are likely to have more links than a purely domestic chain and because different national laws and market practices are involved.

A range of issues arises at the intersection of custody chains and corporate governance. ³¹ Custody chains provide end investors with access to intermediary services that in theory *should* help them to manage participation in general meetings and corporate actions but practical problems to do with the timely flow of communications to and from companies and their end investors through the chain of intermediaries mean that reality sometimes falls short. On the whole, corporate action processes work reasonably well, in particular the cascading of dividends and interests payments down the custody chain, but in spite of the tremendous capabilities of modern technologies, cumbersome mechanisms for communicating voting and related general meeting matters can still persist, especially for retail investors. ³² Investors may find that they have to pay significantly for the privilege of engagement through the charging by their intermediaries of extra fees for voting and other shareholder services. ³³ Investors may also have to take extra steps, such as the lodging of paper powers of attorney, just to establish their eligibility to vote remotely. There can also be more fundamental legal complexities associated with identifying exactly where in the chain control of shareholder rights is located.

In a simple non-intermediated world, an investor in shares acquires directly a bundle of shareholder rights, including voting rights and the right to benefit from corporate actions such as the payment of dividends, stock splits, rights issue offer and takeover offers. If this allocation of rights were transposed directly into the intermediated world, the control of shareholder rights would sit with the end investor. Contractual arrangements along the chain would be designed to achieve this effect where it does not follow automatically from the appliable national corporate and property laws. However, in practice the aggregate impact of the multiple layers of contractual terms on which intermediary services are provided throughout the custody chain can result in end investors having much less control over the exercise of shareholder rights than a direct, non-intermediated investor would have.

⁻

³¹ Eva Micheler, 'Intermediated Securities and Legal Certainty' (2014) LSE Law, Society and Economy Working Papers 3/2014; Jesus Aparicio, 'Enhancing Shareholder Rights in Intermediated Securities Holding Structures across Borders' (2017) 13 New York University Journal of Law & Business 465; Federico Panisi, Ross P Buckley and Douglas Arner, 'Blockchain and Public Companies: A Revolution in Share Ownership Transparency, Proxy Voting and Corporate Governance?' (2019) 2 Stanford Journal of Blcokchain Law and Policy 32.

³² Jill E Fisch, 'Standing Voting Instructions: Empowering the Excluded Retail Investor' (2017) 102 Minnesota Law Review 11.

³³ Law Commission, 'Intermediated Securities - Scoping Paper' (2020) <a href="https://s3-eu-west-2.maganaya.gov/law.com/newd.gov/2020/11/J.gov/Commission.law.com/newd-gov/2020/11/J.g

^{2.}amazonaws.com/lawcom-prod-storage-11jsxou24uy7q/uploads/2020/11/Law-Commission-Intermediated-Securities-Scoping-Paper-1.pdf>, ch 3.

³⁴ Paul Davies, 'Investment Chains and Corporate Governance' in Louise Gullifer and Jennifer Payne (eds), *Intermediation and Beyond* (2019); Law Commission (n 33).

³⁵ Guy Morton, 'Historical Introduction: The Growth of Intermediation and Development of Legal Analysis of Intermediated Securities' in Louise Gullifer and Jennifer Payne (eds), *Intermediation and Beyond* (Hart Publishing 2019); Law Commission (n 33).

Buried in the small print may be terms that relieve or limit intermediaries' responsibilities with respect to the exercise of voting rights by end investors.³⁶ In addition, little-noticed terms of service that permit intermediaries to lend shares may mean that end investors have lost control rights at the relevant time.³⁷ The problems are compounded in the (common) case of international chains due to cross-border variation in both national company laws and the terms on which intermediary services are provided.³⁸ In practice, it is asset managers and proxy advisers that typically play a key role in the exercise of control rights.³⁹

Corporate governance-related problems with custody chains exist in major markets around the world. This article's focus on the situation in Europe, by which is meant the EU/EEA and also the UK, is timely for a number of reasons. First, equity investors in European companies often have no practical alternative to being part of a custody chain because of the way that trading, clearing and settlement is organized. The holding of securities in book entry form will be mandatory for EU/EEA issuers admitted to trading venues by no later than 2023 for new securities and 2025 for existing securities. The UK is now on a different path as regards shares that are still held in paper form but the practical impact of this particular consequence of Brexit is limited because the majority of publicly-traded UK equities are already held in dematerialized form in CREST, the national CSD.

Secondly, the development of a deeper equity culture is a key part of the flagship Capital Markets Union (CMU) project, first launched by the European Commission in 2015 and now in its second five year plan.⁴³ CMU is all about removing barriers to the free

³⁶ Some jurisdictions (notably the US) prohibit or significantly restrict discretionary voting by brokers on behalf of end investors.

³⁷ Hirst and Robertson (n 21); Law Commission (n 21), paras 11.86-11.88. Jaap Winter, 'Cross-Border Voting in Europe' in Klaus Hopt and Eddy Wymeersch (eds), *Capital Markets and Company Law* (OUP 2003).

The EU Securities Financing Transactions Regulation (Regulation (EU) 2015/2365 of the European Parliament and of the Council of 25 November 2015 on transparency of securities financing transactions [2015] OJ L333/1, (onshored in the UK by the European Union (Withdrawal Agreement) Act 2020 (but with certain changes made by statutory instrument)) imposes new disclosure and reporting requirements on fund managers with respect to their use of securities funding transactions to assist investors to understand the risks and returns and make better informed investment choices.

³⁸ What is permissible contractually will always depend on the applicable broader legal framework. Regulatory requirements may require that particular investment products guarantee a right to vote to the end investor: eg, the UK Individual Savings Account Regulations 1998, SI 1998/1870 exemplify.

³⁹ John Kay, 'Review of UK Equity Markets and Long-term Decision Making', para 3.13.

⁴⁰ Daniels (n 20) ((US-focused analysis with the starting premise that 'modern shareholder voting is broken'); Liyuan Meng and others, 'Research on the Application of Blockchain Technology in the Field of Corporate Governance' (2021) 275 E3S Web of Conferences 1064 (China-focused analysis describing AGMs as flawed with respect to information and decision-making functions, and voting procedures).

⁴¹ Regulation (EU) No 909/2014 of the European Parliament and of the Council of 23 July 2014 on improving securities settlement in the European Union and on central securities depositories and amending Directives 98/26/EC and 2014/65/EU and Regulation (EU) No 236/2012 [2014] OJ L257/1 (CSDR), art 3. The recording of securities in book-entry form can take place outside a CSD: CSDR, recital 11. 'Dematerialized form' means that financial instruments exist only as book entry records: CSDR, art 2.1(4).

⁴² CSDR continues to apply in the UK as retained EU law but only in respect of provisions that were operative immediately before exit day, which excludes the dematerialization requirement as it does not apply until 2023 at the earliest. European Union (Withdrawal Agreement) Act 2020, s 3; The Central Securities Depositories (Amendment) (EU Exit) Regulations 2018, SI 2018/1320. Independently of EU law, the UK has instigated its own full dematerialization project for publicly traded shares not held in CREST but at the time of writing it is unclear how vigorously it is being pursued.

⁴³ European Commission, 'Action Plan on Building a Capital Markets Union' COM(2015) 468; European Commission, (n 17).

movement of capital across the single market. The vital role that custody chains play in enabling investors to invest internationally has been mentioned already in this article. That there were significant barriers to cross-border investment associated with post-trade legal requirements and industry practices and processes was first put on the EU policy radar in the early $2000s^{44}$ but it is the Shareholder Rights Directive, first adopted in 2007 and amended in 2017 (now SRD II) that most directly addresses the intersection of custody changes and corporate governance. SRD II, which, as noted earlier, also addresses the stewardship responsibilities of the investment industry through a suite of new transparency obligations on institutional investors, asset managers and proxy advisers, was operative before the end of the Brexit transition and therefore has been implemented in the UK.

The effective go-live date for the SRD II changes relating to custody chains was September 2020. Industry preparations for SRD II, which Broadridge, the US proxy voting and corporate communications services provider, described as 'the most significant initiative for advancing European corporate governance for many years' are receiving their first real tests. This article is therefore among the first to have the benefit of evidence from practical experience to help assess the impact of the requirements.

4. SRD II AND CUSTODY CHAINS – THE NEW REQUIREMENTS

The following table provides a selective summary of the SRD II requirements relating to custody chains. ⁴⁶ In addition to the directive itself, an implementing act lays down supporting minimum requirements for standardized formats, types of information and data elements. ⁴⁷ The use of common formats and message structures is intended to enable efficient and reliable processing and interoperability between intermediaries, companies and shareholders. ⁴⁸ This extends to prescribing the form and content of notices convening general meetings. ⁴⁹ The implementing act also sets deadlines to help ensure the swift processing of transmissions and establishes procedures relating to ensuring the integrity and security of custody chain processes. ⁵⁰ Deadlines depend on the matter in question but in many cases transmission the same business day is required. The development of harmonized market

⁴⁴ Giovannini Group, 'Cross-Border Clearing and Settlement in the European Union' (2001) https://ec.europa.eu/info/publications/giovannini-reports_en; Giovannini Group, 'EU Clearing and Settlement Arrangements' (2003) https://ec.europa.eu/info/publications/giovannini-reports_en; European Commission, 'Clearing and Settlement in the European Union - Main Policy Issues and Future Challenges' COM(2002) 257; European Commission, 'Clearing and Settlement in the European Union - The Way Forward' COM(2004) 312.

⁴⁵ Broadridge, 'SRD II – The Industry State of Readiness' https://www.broadridge.com/intl/article/srd-ii-the-industry-state-of-readiness

⁴⁶ For in depth analysis see the chapters by Matteo Gargantini, Corrado Malberti and Alessio Bartolacelli in Hanne S Birkmose and Konstantinos Sergatis (eds), *The Shareholder Rights Directive II: A Commentary* (Edward Elgar 2021).

⁴⁷ Commission Implementing Regulation (EU) 2018/1212 of 3 September 2018 laying down minimum requirements implementing the provisions of Directive 2007/36/EC of the European Parliament and of the Council as regards shareholder identification, the transmission of information and the facilitation of the exercise of shareholders rights [2018] OJ L223/1 (SRD II IR).

⁴⁸ SRD II IR, rec 2.

⁴⁹ SRD II IR, annex, Table 3.

⁵⁰ SRD II IR, arts 9-10.

standards and the use of modern communication technologies to promote further standardization over and above the prescribed requirements is expressly encouraged.⁵¹

⁻

⁵¹ SRD II IR, recs 3-4 and 11. This work is being taken forward by industry bodies: Market Standards for Shareholder Identification, December 2020 (final version endorsed by industry bodies): https://ecsda.eu/wp-content/uploads/2020/12/2020_12_Market_Standards for Shareholder identification.pdf; Market Standards for General Meetings, 9 September 2020 (final version subject to implementation): https://www.ebf.eu/wp-content/uploads/2017/07/Market-Standards-for-General-Meetings.pdf

SRD II Chapter Ia: provisions with general application	
In-scope companies (COs) (art	Registered office in EEA/UK and shares admitted to
1(1))	trading on EEA/UK regulated market.
In-scope securities (art 1(1))	Voting shares in in-scope companies.
In scope intermediaries (art	Providers of safekeeping, administration or
1(5))	maintenance of securities accounts services (art 2(d)) in so far as: • They provide services with respect to in-scope securities to shareholders (SHs) or other intermediaries. • 3 rd country intermediaries included.
Non-discrimination,	 Intermediaries to disclose publicly applicable charges
proportionality and	for custody chain services, separately for each service.
transparency of costs (art 3d)	 Charges levied on COs, SHs and other intermediaries
	to be non-discriminatory and proportionate to actual
	costs incurred for delivering such services.
	National laws may prohibit intermediaries from
	charging fees for services.
	Chapter Ia: specific requirements
Shareholder identification (art 3a)	 Right for CO to request SH identification from intermediaries (national laws may set minimum % level, not exceeding 0.5%). Intermediary to communicate to CO without delay requested information re SH identity. Where multiple intermediaries, CO requests are to be transmitted along chain without delay; information from intermediaries to be transmitted directly to CO/its nominee. CO can obtain information re SH identity directly from any intermediary in chain that holds the information. National laws may permit CSDs/other intermediaries or service providers to collect information re SHs and to transmit that information to COs. Standardization of minimum requirements re format of requests and information transmitted.
Intermediary to transmit information 'without delay' (art 3b)	 CO→SH/nominee: intermediary to transmit company information re exercise of rights flowing from shares (except where CO sends that information directly to SHs/nominees). SH→CO: intermediary to transmit in accordance with SHs' instructions, information received from SHs re exercise of rights flowing from their shares. Multiple intermediaries in chain: intermediaries to transmit information along the chain (unless it can be directly transmitted to CO or SH/nominee)

	• Standardization of types and formats of information, including security and interoperability.
Intermediary to facilitate exercise of SH rights, including right to participate and vote in general meetings (art 3c)	 Facilitation to be by one of Making arrangements for SH/nominee to be able to exercise rights themselves; or Exercising rights flowing from the shares upon explicit authorisation and instruction of SH and for SH's benefit Electronic confirmation of receipt of votes cast electronically to be sent to person that cast the vote. SH/nominee can obtain, at least upon request, confirmation that their votes have been validly recorded and counted. Confirmations of votes cast sent to intermediaries must be transmitted to SH/nominee without delay either through the intermediary chain or directly. Standardization of formats of confirmations, including security and interoperability

In terms of a high level overview, the SRD II vision is of a single common interface for corporate actions, general meetings and shareholder identification that operates seamlessly across Europe. This vision responds to calls to streamline and simplify shareholder participation and identification stretching back over many years.⁵² To achieve this, there is a need for detailed standardization at the operational level relating to message formatting, communication methods, technical interfaces and so forth. SRD II and the implementing act sensibly recognize that much of this technical work is best left to the market.

Market work on removing the so-called Giovannini barriers to efficient and effective cross-border clearing and settlement has been ongoing since the mid 2000s. The SRD II implementing act acknowledges that voluntary market standards for corporate actions processing were for the most part already applied, and that it was unnecessary therefore to specify more than key elements and principles with respect to those processes. Market-led progress with respect to standardization of general meeting and voting practices had been less successful, however. RRD II therefore has more heavy lifting to do in this area. The most innovative element of the SRD II changes is the introduction of a harmonized mechanism for shareholder identification. Shareholder identification mechanisms already existed in around half of the EEA jurisdictions' national laws but these were not necessarily designed with cross-border investors in mind, supporting processes were not standardized, shareholder optouts were possible in some jurisdictions, and cross-border application of sanctions against

⁵² There is a thread on this topic that can be traced through from the Giovannini Reports (n 44) through to the Report of the High Level Group of Company Law Experts (European Commission 2002) and the Report of the Reflection Group on the Future of EU Company Law (European Commission, 2011). . ⁵³ SRD II IR, recital 11.

⁵⁴ European Securities and Markets Authority, 'Report on Shareholder Identification and Communication Systems, ESMA 31-54-435' https://www.esma.europa.eu/sites/default/files/library/esma31-54-435 report on shareholder identification and communication.pdf>, para 45.

intermediaries (especially third country intermediaries) was patchy.⁵⁵ Shareholder identification therefore represents the area where there are the most challenges – but also opportunities – for the market to provide new standards and systems to meet SRD II compliance requirements.

SRD II is not just about standardizing backroom processes and systems, however. An important advance achieved by SRD II is that it dismantles legal barriers to the cross-border disclosure of shareholder information. However, in the area of harmonization of key concepts on which the smooth and predictable operation of processes and systems depends, it is already clear that SRD II does not go far enough. Of the elements of unharmonized national company laws that continue to act as barriers to seamless cross-border operation of processes relating to general meetings and shareholder identification, none is thornier than the thicket of possible responses to the seemingly simple question: 'who is my shareholder?

Who is the SRD II 'shareholder'?

The EU's vision for a capital market union is maturing but its plans continue to have to skirt around some areas of law that are deeply embedded in national systems and on which there is resistance to harmonization. Aspects of company law and property law are in this category. In 2001 the influential Giovannini Report on cross border clearing and settlement in the EU acknowledged that shares were creations of national legislative regimes and could never completely escape from them, that laws about what securities are and how they may be owned were a basic and intimate part of national legal systems, and that such systems were path-dependent and reflected the local socio-economic culture. That fundamental differences in concepts underlying national laws would be difficult to remove was anticipated. This observation was prescient so far as SRD II is concerned.

The drive towards greater consistency in the operation of custody chains has not yet touched continuing differences in national company laws on whether voting and economic rights are attributed to the first or final layers of holders in a custody chain. ⁵⁸ In other words, whether the 'shareholder', as the person entitled as a matter of company law to assert voting and economic rights directly against the company, means the person at the first layer or at the final layer. The divergence in company laws is closely related to the national legal rules on proprietary interests. In some jurisdictions (such as France), the form for the holding of publicly traded securities is one in which the end investor is regarded as the owner of the securities and the role of the intermediaries in the chain is largely administrative. ⁵⁹ There is also variation within this model, particularly as to whether as a default position the end investor is anonymous vis-à-vis the issuing company (which is the case in France). ⁶⁰ In other jurisdictions, the predominant form is one whereby legal ownership and the direct relationship with the company is at the top of the custody chain and the end investor has only an indirect legal interest that either draws on generic legal concepts such as trusts and

⁵⁵ Ibid, paras 11-23.

⁵⁶ Giovannini Group, 'Cross-Border Clearing and Settlement in the European Union' (n 44) 54-55.

⁵⁷ ibid 60; Art. L. 228-1 Commercial Code.

⁵⁸ ESMA (n 54) annex 1, paras 47-50.

⁵⁹ Victoria Dixon, 'The Legal Nature of Intermediated Securities: An Insurmountable Obstacle to Legal Certainty?' in Louise Gullifer and Jennifer Payne (eds), *Intermediation and Beyond* (Hart Publishing 2019).

⁶⁰ Dixon, ibid. The company could force nominative shares: Art. L. 228-1 Commercial Code; it does not happen in practice for listed issuers

equitable interests (as in the UK and Ireland) or is a bespoke securities interest.⁶¹ In others (such as Belgium and Germany), a co-ownership model as between intermediaries and the end investor is involved.⁶² Moreover, national laws may allow for different answers depending on whether bearer or registered shares are involved.⁶³

In not attempting to harmonize the definition of shareholder, SRD II has left a troubling gap that undermines the aim of the reforms related to custody chains.⁶⁴ The Association for Financial Markets in Europe (AFME) has described the continued existence of twenty seven different definitions of a shareholder across the bloc as 'a significant challenge' for the development of the consistent market standards that are needed to support the operationalization of the new measures on custody chains.⁶⁵ In an earlier report, the European Securities and Markets Authority (ESMA) had anticipated that problems would arise from the lack of agreement on who the law recognises as a shareholder.⁶⁶ In 2020 the European Commission's High Level Forum on the Capital Market Union called for a harmonized definition of 'shareholder' in SRD II as part of a package of measures to foster greater investor engagement and cross-border investment.⁶⁷ The new European Commission Capital Market Union (CMU) action plan picks up this recommendation as a key action.⁶⁸

Consider the implications: if there is uncertainty about whose identity – shareholder or end investor – matters for SRD II purposes, then intermediaries risk contractual and/or legal, regulatory or administrative liability for disclosing sensitive information if they identify the 'wrong' person;⁶⁹ corporate information may not reach the 'right' person at the right time; and votes may be cast wrongly. In effect, improvements at the level of market standards, systems and processes are compromised because more fundamental matters remain uncertain.

Who is the 'end investor'?

Before proceeding to look at how the shareholder/end investor problem is playing out in practice, it is necessary to detour briefly to establish what is meant by the term 'end investor'. This term is not used in SRD II or the implementing act. This article therefore follows market standards in using the term to mean an investor that acquires and holds shares via a custody chain as distinct from an end saver that buys units in pooled funds that then are invested in equities or other securities.⁷⁰ While an end saver is similar to an end investor in having an economic interest in the performance of the equities in which their savings end up, their position is nevertheless distinguishable from that of an end investor who for technical or

⁶¹ Dixon, ibid.

⁶² ibid

⁶³ AGC-EFC, 'Shareholder Rights Directive II Position Paper' 4 August 2020 http://theagc.com/EFC%20SRD%20II%20Position%20Paper%2004-08-20%20FINAL.pdf.

⁶⁴ SRD II, art 2b refers the definition of shareholder back to national laws.

⁶⁵ AFME, Introduction to Shareholder Rights Directive II, March 2020 https://www.afme.eu/Portals/0/DispatchFeaturedImages/Introduction% 20to% 2

https://www.afme.eu/Portals/0/DispatchFeaturedImages/Introduction%20to%20SRD%20II.pdf

⁶⁶ ESMA, (n 54), paras 13-14.

⁶⁷ High Level Forum (n 14),16

⁶⁸ European Commission, (n 17) action 12.

⁶⁹ SRD II, art 3a(6) requires Member States to ensure that intermediaries that disclose shareholder identity information are not considered to be in breach of any restriction on disclosure imposed by contract or by any legislative regulatory or administrative provision, but only where they do so *in accordance with the rules laid down in this Article* (emphasis added).

⁷⁰ Market Standards for Shareholder Identification (n 51); Market Standards for General Meetings (n 51). For further analysis see: Law Commission (n 33).

efficiency-related reasons routes their investment in equities through a custody chain. The end saver in a fund or other collective product is not setting out to invest directly in equities and has no reasonable expectation of getting direct control of the bundle of rights generally associated with being a shareholder. The end saver holds a different type of financial instrument.

The popularity of active and passive funds, pensions, life insurance and other pooled products as the preferred destination for long-term retail savings means that the typical end investor at the end of a custody chain will be an asset manager, pension fund trustee or other institutional investor that itself is subject to an array of investment management duties owed to its end savers. However, it remains possible for the end investor in a custody chain to be a retail or other small direct participant in the capital markets. As noted in an earlier section of this paper, there are signs of a (modest) upturn in direct retail investment in equities in some markets. There is debate about the desirability in principle of retail investor participation in corporate governance given that retail investors are, it is argued, likely to be relatively uninformed and inexpert compared to those in the investment industry. 71 While it is not the purpose here to explore fully the policy issues associated with uninformed voting, this article aligns with others that have not found the risk of uninformed voting to be convincing justification for features of custody chains that legally or practically deprive end investors (retail or, indeed, institutional) of the power to control the exercise of voting or other shareholder rights. ⁷² Market structures that have developed through 'haphazard happenstance and necessity', 73 and that reflect incumbencies and inequalities of bargaining power do not necessarily align with the set of efficient arrangements that are most conducive to improving corporate governance for the benefit of society overall.

Are holders of depository receipts (or even convertible bonds linked to voting shares) to be regarded as end investors? Functionally, since depository receipts are just a convenient mechanism for investors to invest globally in equities, from a corporate governance perspective that is focused on ensuring that control of shareholder rights is located with the person with the appropriate incentives, it could be argued that they should be within the scope of a purposive interpretation of shareholder for SRD II (or alternatively that a DR is a share for this purpose). On the other hand, a depository receipt is a different security from a share and it could be argued that it stretches the concept of shareholder too far to include DR holders. The relationship between the investor, the depository bank and the custodian in the issuer's home country is not the standard custody relationship around which SRD II is designed. On this issue, market standards envisage the chain going only as far as the intermediary of the depository bank where the underlying shares are held⁷⁴ but Member

_

⁷¹ Morton (n 35). But investment intermediaries often have limited incentives to become better informed, so query where the line might be drawn if being informed became a criterion for eligibility to vote as a shareholder: Marcel Kahan and Edward B Rock, 'Index Funds and Corporate Governance: Let Shareholders Be Shareholders' (2020) 100 Boston University Law Review 1771. Dorothy S Lund, 'The Case Against Passive Shareholder Voting' (2018) 43 Journal of Corporation Law 493. Lucian Bebchuk and Scott Hirst, 'Index Funds and the Future of Corporate Governance: Theory, Evidence and Policy' (2019) 119 Columbia Law Review 2029.

⁷² Fisch (n 32).

⁷³ Daniels (n 20), 406.

⁷⁴ Market Standards for Shareholder Identification (n 51).

States have taken different views on the treatment of depository receipts (and convertible bonds) in their implementation of SRD II.⁷⁵

Both the preceding and following section of this article explain why the SRD II definition of shareholder requires further clarification. The takeaway from this section is the supplementary point that such reform will need to be comprehensive to prevent similar uncertainties and inconsistences to those around 'who is my shareholder?' persisting at the level of 'who is my end investor?'.

SRD II: illustrating the limited/troubled application where the shareholder is not the end investor

We return now to the problematic application of SRD II in circumstances where as a matter of national company law the shareholder is not the end investor. A paper published by the Association of Global Custodians - European Focus Committee (AGC-EFC) in August 2020 revealed a disturbingly chaotic picture at the national level on this point: some Member States company laws take a different position on whether the shareholder and end investor are one and the same depending on whether the shares are held in registered or bearer form; some Member States define the shareholder as the person at the top of the chain in their company laws but in the implementation of SRD II have sought to take a purposive approach that extends the scope to the end investor, which may not be legally defensible; in implementing SRD II some have gone down to the level of the end investor for the purposes of the shareholder identification right but not necessarily for other SRD II custody chains requirements, an approach which again is of doubtful legal validity. SRD II thus struggles on two levels: it has created new uncertainties within national legal systems; and it has manifestly failed to achieve cross-border harmonization on a pivotal concept.

The cleanest approach from a national legal certainty is simply not to go beyond the top layer where that person is identified as the shareholder as a matter of company law. The UK exemplifies this approach. But while this approach avoids creating uncertainty within national law, it does not align with the harmonizing spirit of SRD II.

The UK is a registered shares jurisdiction. The register, which for dematerialized shares held through CREST is maintained by Euroclear UK and International (EUI), records legal ownership⁷⁷ and is prima facie evidence of matters required to be recorded.⁷⁸ The inclusion on the register of details of underlying beneficial interests is not permitted.⁷⁹ The identity of the end investor (also described as the beneficial owner) is not generally known to the issuing company and it is only the person named on the register (the 'member') that is formally entitled to raise shareholder rights against the company. In implementing SRD II, the UK proceeded on the basis that the SRD II shareholder identification and custody chain changes were largely irrelevant because as a 'name on register' jurisdiction, companies already knew the identity of their (registered) shareholders and communications with

⁷⁷ The Uncertificated Securities Regulations 2001, SI 2001/3755 (USR), reg 20.

⁷⁹ USR, reg 23(3) and reg 40(3).

⁷⁵ AGC-EFC (n 63).

⁷⁶ ibid.

⁷⁸ USR, reg 24. The part of the register with respect to shares still in materialized form is maintained by the company. In its dematerialization project the UK is considering a model whereby a separate part of the electronic register would be maintained for dematerialized shares not held in CREST.

shareholders took place directly. This approach is supported by a recital in the amending directive that is consolidated as SRD II, which states expressly that the directive 'does not affect the beneficial owners of other persons who are not shareholders under the applicable national law'. ⁸⁰ But it is surprising outcome when set against the general purpose of SRD II to improve the operation of custody chains, and the approach of the implementing act. As such, it has attracted criticism. ⁸¹

It is not that the UK is diametrically opposed to the principle of companies being able to identify their end investors and of the transmission of information up and down custody chains and the facilitation of end investor control over the exercise of shareholder rights. Notwithstanding the firm adherence to the 'name on register' system and the fierce defence of it by market practitioners who view it as being critical to the delivery of the legal certainty on which the capital markets depend, since even before SRD II (though partly under the influence of earlier EU law) company law in the UK has contained a number of provisions that are designed to enable companies to identify end investors (beneficial owners) and also to foster communication and engagement between companies and end investors. Public companies can inquire into the identity of their end investors through the use of statutory powers to extract information from persons whom the company knows or has reasonable cause to believe to be interested in its shares. 82 It is permissible for a company to include provision in its articles enabling a registered shareholder to nominate another person to exercise shareholder rights. 83 A registered shareholder may also nominate other persons to enjoy corporate information rights.⁸⁴ A registered shareholder that holds shares on behalf of others can exercise voting rights attached to different shares in different ways. 85 The casting of votes on shares held in a nominee account in accordance with the wishes of the end investors can be facilitated by the nominee registered shareholder appointing those end investors as its multiple proxies.⁸⁶ It is even permissible for end investors to count towards the minimum number/percentage of registered shareholders required for certain members' requests.87

There are mixed views on how well these various provisions work in practice.⁸⁸ Hence the disappointment in some quarters that SRD II implementation was not used as the opportunity to put in place a stronger system for end investors to have their say in corporate governance. The point is also made that it is likely that market participants that are active in multiple markets will still want to align UK practices with to those mandated by SRD II given the significant investment in the development of systems to support the SRD II requirements and the inefficiencies of maintaining idiosyncratic local market practices.⁸⁹

80 Directive 2017/828, recital 13.

⁸¹ Dermot Turing, Clearing and Settlement (3rd edn, Bloomsbury Pofessional 2021), 642-644...

⁸² Companies Act 2006, s 793.

⁸³ Companies Act 2006, s 145.

⁸⁴ Companies Act 2006, s 146. This applies to companies with shares admitted to trading on a UK or EU regulated market.

⁸⁵ Companies Act 2006, s 152.

⁸⁶ Companies Act 2006, s 324.

⁸⁷ Companies Act 2006, s 153.

⁸⁸ The European Commission described the UK's domestic rule that allows companies to identify their end investors (Companies Act 2006, s 793) as 'successful' and 'effective' but also 'highly intensive and time-consuming': European Commission, Proposed SRD II Impact Assessment SWD(2014) 127, 34
⁸⁹ Turing (n 81).

However, the possibilities for convergence at the level of market standards, systems and processes are bounded by the national legal requirements and thus the retention at least to some extent of parallel processes may be unavoidable.

Ireland, on the other hand, exemplifies the legal uncertainties that arise from adopting a more purposive approach to SRD II implementation. Ireland, which is also a registered shares jurisdiction, has run into some difficulties by implementing SRD II in a way that seeks to cater for both 'name on register' shareholders and end investors. The term 'shareholder' does not have a single statutory definition in Irish company law and the regulations implementing SRD II refer equally to the end investor and the registered holder. Practitioners report that this has led to market uncertainty about exactly who in the custody chain is within the scope of the new requirements. The implications of such uncertainty are significant. The Association of Global Custodian-European Focus Committee notes that end investors would, in the end, be reliant on the Irish courts to protect them on the basis of a purposive interpretation of the term 'shareholder'. This unsatisfactory position is inimical to legal and commercial certainty. Euroclear Bank, which is the post-Brexit CSD for the Irish equities market, has adopted the 'name on register' approach in determining the 'shareholder' for SRD II purposes.

The significance of the new right for companies to identify their shareholders

It is one thing to know your shareholder but not your end investor. Not even knowing your shareholder is a different problem. Whereas the former problem is an unintended (but not wholly unforeseen) consequence of not addressing a delicate company/property law matter, the latter problem is the one that the SRD II shareholder identification right was always meant to tackle. Recognising that shareholder anonymity vis-à-vis the company is the common default position across swathes of the European capital markets, especially those jurisdictions where shares continue to be held mainly in bearer form is thus key to appreciating the significance of the SRD II shareholder identification mechanism.

A 2016 study by the European Central Securities Depositories Association (ECSDA) found that registration with respect to shares held in CSDs was mandatory in roughly half of the 38 European markets surveyed. ⁹⁴ Countries in the mandatory group included the UK, Ireland, Denmark, Finland, Sweden and Norway, and such bearer shares as did exist in these countries were effectively a legacy effect and represented only a small proportion of the markets. The states where registration was still optional included Germany, France and Luxembourg and in these markets, bearer shares continued to be widely used, at the level of 50-75 per cent, or even up to 100 per cent, of shares held within CSDs. (Note that while the term 'bearer' shares has connotations of physical certificates of ownership, it continues to be

⁹⁰ European Union (Shareholder Rights) Regulations 2020, SI 2020/81 (Ireland).

⁹¹ Mason, Hayes & Curran, 'Shareholder Rights Directive II – Identification of Shareholders and Facilitating Shareholders' Rights' (24 June 2020) https://www.mhc.ie/latest/insights/shareholder-rights-directive-ii-identification-of-shareholders-and-facilitating-shareholders-rights

⁹² AGC-EFC (n 63).

⁹³ ECB Advisory Group on Market Infrastructures for Securities and Collateral (AMI-SeCo), 'Corporate Events Compliance Report: 2021 Monitoring Exercise' (December 2021) 16-17.

⁹⁴ ECSDA, The Registration of Securities Holders, 19 July 2016, https://ecsda.eu/wp-content/uploads/2016_07_19_ECSDA_Registration_Report.pdf

used in the dematerialized securities world to describe shares whose owners' names are not recorded in an official register.)

France exemplifies the standard default position with respect to dematerialized bearer shares in which companies do not know who their shareholders and end investors (if different) are. As an illustration, Air France-KLM explains how the French dematerialized bearer share system works as follows:

In France and the Netherlands, holding shares in bearer form is the most common method: the purchase and day-to-day management is entrusted to the financial intermediary chosen by the shareholder. This method of holding shares enables the different shares in a portfolio to be regrouped with a single intermediary. The financial intermediary alone knows the identity of the shareholder.⁹⁵

Other French companies' websites provide similar explanations.⁹⁶

Bearer shares whose holders are not known to the issuing company are also common, for example, in Germany. Market sentiment is that SRD II will have a massive impact on most German public companies.⁹⁷

Before SRD II, French law already provided a mechanism (*titre au porteur identifiable* (TPI)) whereby companies could discover the identities of shareholders. There were also shareholder identification procedures under German domestic law. The TPI procedure was introduced into French law in 1987 but through to 2019 (when the SRD II changes tool effect), this was possible only if this was provided for in the articles of association. The TPI system was further refined in 2001 to facilitate its use with respect to third country intermediaries holding securities for clients resident outside France (providing a kind of template for SRD II's treatment of third country intermediaries). Overall, this system was successful but it was considered to be expensive for issuers so they did not do it very often, and usually only ahead of a general meeting (especially if contentious).

Unsurprisingly, as well as changing the law, SRD II has galvanized market developments aimed at providing technologies to simplify shareholder identification processes and ensure SRD II compliance throughout the chain. ⁹⁹ Service enhancements developed in response to the revised regulatory environment are expected to enable shareholder identification with more transparency and lower cost than existing procedures. ¹⁰⁰ Unfortunately, it appears that some legacy national systems remain in operation in parallel to

⁹⁷ K Pohl, 'How SRD II Fundamentally Changes Shareholder ID for Many European Issuers' *IR Magazine* https://www.irmagazine.com/shareholder-targeting-id/how-srd-ii-fundamentally-changes-shareholder-id-many-european-issuers.

⁹⁵ https://www.airfranceklm.com/en/finance/shareholders/being-shareholder

⁹⁶ https://www.spie.com/en/finance/becoming-shareholder

⁹⁸ Art L 228-2 Commercial Code. For securities admitted to trading on a regulated market, any clause to the contrary in articles of association is now overriden

⁹⁹ Eg Investorinsight, a new shareholder identification service launched by ESES CSDs (Euroclear France, Euroclear Belgium and Euroclear Nederland): https://www.euroclear.com/newsandinsights/en/press/2021/2021-mr-10-euroclear-rolls-out-new-shareholder-identification-service.html; German CSD response: https://www.clearstream.com/clearstream-en/products-and-services/asset-services/c20019-1991462. https://www.clearstream-en/products-and-services/asset-services/c20019-1991462. https://www.clearstream-en/products-and-services/asset-services/c20019-1991462. https://www.clearstream-en/products-and-services/asset-services/c20019-1991462.

those adopted for the purposes of SRD II compliance. ¹⁰¹ The retention of parallel systems is not conducive to the achievement of the aim of a single common interface.

Operational challenges and opportunities

The operational challenges and opportunities arising from SRD II come into focus at this point.

In 2017 ESMA presented a general assessment of the level of harmonization of national regulatory frameworks for shareholder identification and communication across the EEA. ¹⁰² Its findings indicated low levels of harmonization in many operational areas: top down corporate communication processes through chains of intermediaries were diverse and based either on national practices or legal frameworks; standard forms and formats for bottom up communication between issuers and shareholders were available in several jurisdictions but with non-harmonized contents, and intermediaries seemed to play a limited role in many areas. The ESMA survey fed into the SRD II implementing act (SRD II IR) and provides useful background context against which to consider its operational impact.

In short, to meet the SRD II IR requirements for information to be in electronic and machine-readable format using internationally applied industry standards (ie ISO20022) has required a quite significant technology infrastructure upgrade. This has not been without glitches. 103 There have been reports of technological problems due to internal systems not being ready in time for new standardized electronic messaging formats, ¹⁰⁴ and of interim solutions having to be deployed. 105 In its December 2021 Report, the ECB's Advisory Group on Market Infrastructures for Securities and Collateral (AMI-SeCo), which has responsibility for compliance monitoring against the market standards reported challenges in respect of cross-border activity owing to differences in the definition of shareholder and operational procedures (among other matters), problems related to some intermediaries not being able to process information in the correct electronic format, difficulties in verifying that shareholder identification requests were really coming from or on behalf of issuers because not all parties in the chain were in contractual relationships with each other, and issues related to the continued use of paper-based requests. 106 AMI-SeCo actually found a dip in compliance with the market standards on corporate actions, notwithstanding that this was already the most mature area pre-SRD II, but acknowledged that this could be related to enhanced rigour in the monitoring process, adjustments to the assessment methodology and the introduction of a new facility for non-domestic markets to report non-compliance cases for the first time. In

1

 ¹⁰¹ Edoardo D Martino and Alessio M Pacces, 'Shareholders Right Directive II: The Italian Implementation' (2020) 2 Revue Trimestrielle de Droit Financier 43.
 102 ESMA (n 54).

¹⁰³ Tim Human, 'Europe's New Shareholder Disclosure Regime Hit by Delays and Tech Issues' [2021] *IR Magazine* 25 January https://www.irmagazine.com/shareholder-targeting-id/europes-new-shareholder-disclosure-regime-hit-delays-and-tech-issues; Marcello Topa, 'SRD II - Ironing Out the Creases' (*Global Investor Group*) 13 May 2021 https://www.globalinvestorgroup.com/articles/3696714/srd-ii-ironing-out-the-creases; State Street, 'Over the First SRD II Hurdle: In Retrospect' (*2021*)

https://www.statestreet.com/content/dam/statestreet/documents/Articles/over-the-first-srd-II-hurdle-in-retrospect-global-issue.pdf.

¹⁰⁴ State Street, ibid (market standard ISO 20022 delayed to August 2020, with knock on effects for CSDs (Euroclear and Clearstream) and intermediaries (Broadridge)).

¹⁰⁵ eg, by Clearstream (German CSD): https://www.clearstream.com/clearstream-en/products-and-services/asset-services/c20019-1991462. State Street ibid (issuer agents still using pre- SRd II process flows).
https://www.clearstream.com/clearstream-en/products-and-services/asset-services/c20019-1991462. State Street ibid (issuer agents still using pre- SRd II process flows).
https://www.clearstream.com/clearstream-en/products-and-services/c20019-1991462. State Street ibid (issuer agents still using pre- SRd II process flows).

total, out of 40 markets. six were found to be fully compliant with the market corporate actions standards, 15 with the T2S corporate actions standards and six with the shareholder identification standards

On the more positive side, it can also be observed that SRD II is driving technological innovation and creating new business opportunities for both incumbents and new entrants. For instance, the (I)CSD group Euroclear launched a new InvestorInsight service aimed specifically at supporting intermediaries in meeting their SRD II obligations. ¹⁰⁷ The other major European (I)CSD group Clearstream saw the transition to ISO20022 as an opportunity to encourage its clients to sign up to the enhanced proxy voting service provided by ISS, its sister company. 108 Upcoming regulatory changes (ie SRD II) played a part in positioning Proxymity, an investor communications and proxy voting services start-up initially sponsored by Citi, to attract some of the world's largest financial institutions to invest in the platform and establish the company as an independent business. 109 Broadridge identified itself as having played a key role in helping tier-one custodians, banks and brokers comply with SRD II requirements, noted 'unprecedented' demand for its solutions, stressed its ability to provide same-day responses, and highlighted the capacity of its platform to leverage advanced APIand blockchain-based technologies for timely, secure and compliant response to shareholder disclosure requests. 110 BNY Mellon, the largest global custodian, was among those that worked with Broadridge to enhance the proxy voting services offered to clients in response to SRD II changes.¹¹¹ Other examples of innovation could be given but this small sample can suffice to give a flavour of the market response to SRD II as both a challenge and an opportunity.

This prompts the question: 'is technology today – or in the very near future – poised to solve the barriers to effective shareholder/end investor engagement created by custody chains?' But before turning to that topic, this section concludes with brief discussion of two remaining issues that have surfaced as points of concern in the first phase of SRD II implementation.

Corporate governance implications of giving companies the right to know their shareholders earlier

While it is the case that SRD II formally provides that shareholder personal data can only be processed in order to identify shareholders in order, in turn, to communicate directly with a view to facilitating the exercise of shareholder rights and shareholder engagement with the company, disentangling mixed purposes on the part of companies requesting shareholder identification would likely to be difficult if not downright impossible. Companies do not need to give reasons for triggering shareholder identification under SRD II and a 'real'

¹⁰⁷ https://www.euroclear.com/campaigns/en/innovation/InvestorInsight/Intermediaries.html

¹⁰⁸ https://www.clearstream.com/clearstream-en/products-and-services/asset-services/c20019-1991462

¹⁰⁹ https://www.uktechnews.info/2020/05/05/proxymity-secures-16-52-million-investment-from-bny-mellon-citi-clearstream-computershare-deutsche-bank-hsbc-jp-morgan-and-state-street/

¹¹⁰ https://www.broadridge.com/intl/article/rising-to-the-challenge-of-srd-ii

https://www.bnymellon.com/content/dam/bnymellon/documents/pdf/emea/srd-ii-regulatory-programme-update-february-2020.pdf.coredownload.pdf
112 SRD II. art 3a(4).

¹¹³ The comment that the qualifications in art 3a(4) amount to 'little more than wishful thinking' overall is apt: Matteo Gargantini, 'Article 3A: Identification of Shareholders' in Hanne S Birkmose and Komstantinos Sergakis (eds), *The Shareholder Rights Directive II: A Commentary* (Edward Elgar 2021) 46.

purpose of using it as a tool simply to make an early discovery of who is investing in a company's shares could be easily masked.

There is a rich academic and practitioner literature exploring whether the tactical building up of 'hidden' stakes by activist and other investors before they hit identification/disclosure thresholds is beneficial to corporate governance overall. In that regard, it is notable that few Member States have taken advantage of the option to set a minimum level of shareholding threshold (not exceeding 0.5 per cent) on the right for companies to identify their shareholders. This option was inserted into SRD II at a last stage in the legislative process and as a compromise between the interests of investors that would prefer to remain anonymous and the company's interest in knowing who has invested in its shares. By not opting in, Member States have sent quite a strong signal as to the perceived public policy value of this compromise.

The impact on activism of the new SRD II shareholder identification right is somewhat attenuated by the fact that the enables the company, rather than the market as a whole, to identify shareholders. However, Member States could go further in the direction of market transparency in their national implementation: Italy exemplifies, although it is also worth noting that Italy is also one of the Member States that has opted into the 0.5 per cent threshold. It has been argued that from a governance perspective the interaction of existing Italian laws and its implementation of the SRD shareholder identification right could create more problems than it solves. It

Information rights and facilitated exercise of shareholder rights – are opting-out and outsourcing still allowed?

The European Commission's conviction that the proposals that were eventually adopted in SRD II were needed was based on wide consultation across the sector that identified recurring problems that in general made the cross-border exercise of rights flowing from securities by investors both difficult and costly. Contractual terms and practical realities of the way in which custody chains operate can sometimes confound the expectations of smaller and retail investors that want to have a say in the way that companies are run. While institutional investors are more adept at navigating these complexities, even they can encounter some difficulties. 119

Nevertheless, for many end investors the delegation of voting discretion to other finance professionals may represent the rational choice. Many functions in the financial markets are outsourced to specialists, a practice which allows for a concentration of expertise,

¹¹⁴ Recent articles discussing the potentially significant impact on the corporate governance landscape of legal and/or technology (blockchain) changes that make the identity of end investors more visible: Platt (n 19); David Yermack, 'Corporate Governance and Blockchains' (2017) 21 Review of Finance 7.

¹¹⁵ Data published by ESMA: https://www.esma.europa.eu/press-news/esma-news/esma-publishes-list-thresholds-shareholder-identification

¹¹⁶ Martino and Pacces (n 101).

¹¹⁷ ibid.

¹¹⁸ European Commission, 'Proposal for a Directive of the European Parliament and of the Council Amending Directive 2007/36/EC as Regards the Encouragement of Long-Term Shareholder Engagement and Directive 2013/34/EU as Regards Certain Elements of the Corporate Governance Statement' COM(2014) 213, 5-6. See also the accompanying Impact Assessment SWD(2014) 127, 35-36.

¹¹⁹ Law Commission (n 33).

¹²⁰ Law Commission (n 21) paras 5.84-5.88; Twemlow (n 30).

economies of scale and efficient management of cross-border diversified portfolios.¹²¹ In some case delegation may be a function of regulatory requirements that make managing investors for others a regulated activity requiring specific authorization.¹²² The prevalence within custody chains of provisions that manage the allocation of corporate governance control rights through contract speaks powerfully to their value and popularity as an efficient and cost-effective way for end investors, including institutional investors and small funds, to structure their investment activity.

The possibility of shareholders' opting out from services designed for their benefit is not expressly excluded by SRD II and there is an in-passing reference to 'unless otherwise agreed by the shareholder' in one provision of the implementing act. ¹²³ However, the position on this important point could be made clearer. In a future reform of SRD II, the ability of shareholders to waive rights that exist for their benefit should be put beyond doubt.

By the same token, shareholder opt-outs do not (and should not in future) extend to the new corporate rights with respect to shareholder identification: shareholders cannot block intermediaries from providing the required information if a company requests identification because this measure is not purely pro-shareholder.

The post-SRD II revised market standards for general meetings stress that opt outs must be at the initiative of the investor, that intermediary opt-out is not permitted, and that investors must be able to make informed decisions based on clear explanations as to the default position being that they will be offered the services, the price of the services and the implications of opting out. There is a general principle in the standards to the effect that cost should not influence the decision to opt out but since the cost saving involved in opting out will likely always matter to investors, this principle is presumably intended to be related to the obligation on intermediaries to offer services at costs which are appropriate, proportionate and non-discriminatory.

It also remains possible for a shareholder to appoint a nominee to exercise their rights and/or for an intermediary to exercise these rights on their behalf provided the intermediary does so upon express authorization and instruction of the shareholder and for the shareholder's benefit. The requirement for 'express authorization and instruction' may help to address the problem of wording that limits an investor's control being buried in the contractual small print.

5. WILL TECHNOLOGY ELIMINATE CUSTODY CHAINS? NOT QUITE YET ...

The Giovannini Group's forward looking assessment in the early 2000s was that the removal of barriers to efficient cross-border clearing and settlement would continue to be best achieved through a combination of private sector initiative and public sector intervention. ¹²⁶ That insight remains as convincing today as it was twenty years ago.

23

¹²¹ Morton (n 35); Kay (n 39) para 3.9.

¹²² Law Commission (n 21) paras 4.13 and 6.19.

¹²³ SRD II IR, art 2(4).

¹²⁴ Market Standards for General Meetings (n 51).

¹²⁵ SRD II, art 3(1).

¹²⁶ ibid, 59-60.

There is intense market, regulatory and academic interest in the potentiality of digital technologies to transform post-trade processes. ¹²⁷ The use of new technologies in the areas of shareholder identification, corporate actions and general meetings is already growing. ¹²⁸ APIs – interfaces that enable interaction, access and data exchange between different software systems – are especially apt for the purpose of facilitating easier information flows up and down custody chains. As noted earlier, Broadridge has made much of its API capabilities in the context of its SRD II services. Alongside this, DLT – in simplest terms, a tamperproof recordkeeping service that anyone can access ¹²⁹ - has an obvious potential application in the modernization of the way in which corporate registers and other records are maintained and shared. Models for DLT-based utility-type services for a decentralized directory of information to identify end investors and for electronic voting are already in development. ¹³⁰

The more profoundly transformative potential of digital technologies in the world of post-trade is also starting to take shape. The outline of a future world in which technology has eliminated custody chains by establishing a different norm whereby securities are held and settled in the names of end investors can be discerned. The Swiss Stock Exchange Group (SIX) is playing a pioneering role in this respect with the first tokenised bond issuance using DLT technology on its new SIX Digital Exchange in September 2021 and strong predictions of more to follow at pace. However, there are many factors that explain why 'grand visions of disintermediation or digital transformation at scale are still far from being realized'. Significant restructuring is required and business cases for this are uncertain. Delivery versus payment settlement design in the DLT environment, including the need (or not (ie, if settlement is instantaneous)) for liquidity provision and access to central bank money is

¹²⁷ Daniels (n 20), 406; Panisi, Buckley and Arner (n 31). Discussing this as an element of the benefits that transformative technology could bring to the financial markets generally: Emilios Avgouleas and Aggeios Kiayias, 'The Promise of Blockchain Technology for Global Securities and Derivatives Markets: The New Financial Ecosystem and the "Holy Grail" of Systemic Risk Containment' (2019) 20 Journal of Business Organization Law Review 81. And examining improvements in voting as part of a wider survey of potential impact of blockchain on corporate governance: Yermack (n 114). On the regulatory side: ECB Advisory Groups on Market Infrastructures for Securities and Collateral and for Payments, 'The Use of DLT in Post-Trade Processes' (2021).

¹²⁸ Daniels (n 20) 434 (describing a number of examples including pilot programmes by NASDAQ to use blockchain-based shareholder e-voting systems on Estonia's stock exchange and in South Africa). Also Panisi, Buckley and Arner (n 31); Avgouleas and Kiayias (n 123); the examples of SRD II-related innovation mentioned in XXX above.

¹²⁹ Fully decentralized open architecture blockchains that anyone can access are one possibility but a 'permissioned' blockchain run by a trusted central authority has been identified as being more likely in the securities context: ECB Advisory Group on Market Infrastructures for Securities and Collateral, 'The Potential Impact of DLTs on Securities Post-Trading Harmonisation and on the Wider EU Financial Market Integration' (2017), 46.

¹³⁰ ECB Advisory Group on Market Infrastructures for Securities and Collateral, 'Potentiona Use Cases for Innoivative Technologies in Securuties Post-trading (2019), ch 3 and ch 5.

¹³¹ Ian Allison, A National Stock Exchange That Supports NFTs? Welcome to Switzerland' 29 November 2021, https://www.coindesk.com/business/2021/11/29/a-national-stock-exchange-that-supports-nfts-welcome-to-switzerland/ reporting an interview with the SDX Chair, Thomas Zeeb in which Zeeb claims that the floodgates are now open and looks forward to 'an acceleration of some of the pent-up stuff we've been keeping a lid on for a while'.

World Economic Forum, 'Digital Assets, Distributed Ledger Technology and the Future of Capital Markets'
 (2021) https://www3.weforum.org/docs/WEF_Digital_Assets_Distributed_Ledger_Technology_2021.pdf;
 Dirk A Zetzsche, Ross P Buckley and Douglas W Arner, 'The Distributed Liability of Distributed Ledgers:
 Legal Risks of Blockchain' [2018] University of Illinois Law Review 1361.
 World Economic Forum, ibid.

currently underexplored.¹³⁴ Regulatory choices regarding which barriers to remove and what restrictions to impose in order to ensure a safe and efficient digital environment that does not create new prudential and/or conduct risks and maintains regulatory neutrality between different types of provider is another attention point.¹³⁵ The potential extension of the settlement finality protection afforded to established payment systems and securities settlement systems is an important legal topic.¹³⁶ Interoperability both between legacy and new DLT systems and also between DLT system is another key concern.¹³⁷ So, too is, DLT governance.¹³⁸ EU and UK bodies clearly recognise that facilitating greater use of digital technologies and ensuring their resilience and safety are two sides of the same public policy coin.¹³⁹

Besides, as discussed earlier, settlement services are typically just one part of a range of asset servicing and collateral management services offered by financial market infrastructures and their custodian/intermediary participants. This whole ecosystem needs to be redesigned to ensure that the rich diversity of investor services and economies of scale for all parties are not compromised by the adoption of new technology. There is also the consideration that investor preferences for direct recording in the names of end investors in place of the established arrangements are largely untested (some may prefer a default position of being anonymous vis-à-vis the company, even if that is subject to the company's right to seek identification). Moreover, there are also broader issues as stake with respect to new digital technologies, such as whether they are truly tamperproof, their propensity to be used for criminal activity, and their implications for privacy and data protection. He is prospect of multijurisdictional legal upheaval associated with the transition from a world of traditional financial instruments held in dematerialized electronic form to native digital assets.

-

¹³⁴ ibid, 27. Also Andrea Pinna and Wiebe Ruttenberg, 'Distributed Ledger Technologies in Securities Post-Trading: Revolution or Evolution?' 172 https://www.ecb.europa.eu/pub/pdf/scpops/ecbop172.en.pdf ('. Neither market players nor DLT initiatives have given significant attention to the question of how the cash leg of a trade could be linked to the securities leg'). But this is an area that is moving fast: 'Banque de France conducts a new Central Bank Digital Currency experiment', press release 21 June 2021 https://www.banque-france.fr/sites/default/files/medias/documents/cp_mnbc_seba_2021-06-21_en.pdf (detailing an experiment to use central bank digital currency issuance on a blockchain to simulate the settlement of listed securities and thus trigger their delivery in T2S).

¹³⁵ Randy Priem, 'Distributed Ledger Technology for Securities Clearing and Settlement: Benefits, Risks and Regulatory Implications' (2020) 6 Financial Innovation 11.

¹³⁶ In the EU, the legal protections for settlement finality derive from Directive 98/26/EC of the European Parliament and of the Council of 19 May 1998 on settlement finality in payment and securities settlement systems [1998] OL L166/45. The possibility of extension to DLT ledgers is already on the table: European Commission, Proposal for a Regulation of the European Parliament and of the Council on a pilot regime for market infrastructures based on distributed ledger technology (COM)2020 594. Switzerland has already taken this step: Federal Act on the Adaptation of Federal Law to Developments in Distributed Electronic Register Technology, in force from 1 August 2021.

¹³⁷ ECB Advisory Groups on Market Infrastructures for Securities and Collateral and for Payments (n 127), 2. ¹³⁸ Ibid.

¹³⁹ Proposal for a Regulation of the European Parliament and of the Council on digital operational resilience for the financial sector and amending Regulations (EC) No 1060/2009, (EU) No 648/2012, (EU) No 600/2014 and (EU) No 909/2014 COM/2020/595. Operational resilience is also a policy priority for the Bank of England – see the suite of measures relating to FMIs accessible at https://www.bankofengland.co.uk/paper/2021/bank-of-england-policy-on-operational-resilience-of-fmis.

¹⁴⁰ Panisi, Buckley and Arner (n 31); Priem (n 135).

All of which points more towards a process of incremental change rather than sudden transformation. A period of step-by-step transition, when new tools and actors compete with tried, tested and trusted existing business models and incumbents is to be expected. This period is likely to be characterized by experimentation facilitated, where necessary, by 'regulatory sandbox-style' exemptions from normal regulatory requirements. Some envisage this transition period lasting up to 30 years, although such a long timeframe feels extremely conversative given the pace at which breakthroughs are being announced. Although of the touted benefits will actually materialize – some will likely prove to be hype.

The conclusion that the custody chain as we know it today is therefore not about to disappear makes the need to address the shortcomings of SRD II an all the more urgent policy priority. There is the prospect of a virtuous circle whereby reform of SRD II both enables the exploitation of current technologies to their fullest extent and drives further market innovation.

The standout issue for the EU legislative bodies is to resolve the uncertainty about who is meant by the 'shareholder' in SRD II. This, of course, needs to be tackled in a way that does not endanger the legal certainty with respect to proprietary and security interests that underpins market systems that, on a daily basis, safely process an eye-wateringly large volume and number of transactions. ¹⁴⁶ The review of this issue also needs to be mindful that change will cut both ways: improving the efficiency of end investor participation in corporate governance but at the same time also providing what in practice is a new tool for the deanonymization of the end investor vis-à-vis the company. The distinction between the end investor and end saver, and the position for SRD II purposes of holders of depository receipts and convertible bond holders are other matters that should also be clarified when the opportunity arises.

There are other aspects of the legal and regulatory framework around shareholder voting and corporate actions that would also benefit from further harmonization to complement technological developments and support the further evolution of standardized market practices. The original version of what is now SRD II made a start with respect to harmonizing the arrangements for general meetings of shareholder meetings but significant differences in practice remain. Record dates is one topic that comes up frequently as a candidate for further harmonization, and it is on the European Commission's radar. While there is already a harmonized requirement for companies to set record dates, there is a

26

¹⁴¹ Priem (n 135); Carlo RW de Meijer, 'The Roles of CSDs in a Blockchain Environment' (2020) 12 Journal of Securities Operations & Custody 167.

¹⁴² eg, European Commission (n 136).

¹⁴³ Priem (n 135), citing work of the German Banking Industry Committee.

¹⁴⁴ See, eg, the Swiss developments discussed by Allison (n 131). Examples will continue to multiply. Just as this paper was being finalised, the Financial Times was reporting the latest inroads of technology into forex transactions, namely a bilateral arrangement between HSBC and Wells Fargo to settle forex transactions directly on blockchain, thereby cutting out the existing market infrastructure: Eva Szalay and Philip Stafford, 'HSBC and Wells Fargo to Settle Currency Trades with Blockchain' 13 December 2021.

¹⁴⁵ Edmund Schuster, 'Cloud Crypto Land' (2021) 84 Modern Law Review 974.

¹⁴⁶ European Commission (n 17), annex, 7 (particular attention to be paid to effectiveness and efficiency considerations, as well as legal certainty aspects).

¹⁴⁷ High Level Forum (n 14) 16.

¹⁴⁸ European Commission (n 17), annex, 7.

¹⁴⁹ SRD II, art 7(2).

certain amount of divergence between Member States in the calculation of the date resulting in what ESMA has described as only 'a medium level of convergence'. A process that allows the determination of voting entitlements to be carried out with certainty and pan-European consistency is essential if the goal of ensuring that shareholder engagement is not distorted by procedural/technical issues is to be achieved. Moreover, in some countries, shareholders still have to lodge a paper power of attorney as part of the general meeting remote voting process and/or companies are obliged/permitted to communicate with their shareholders via a range of channels including traditional post. Legacy requirements such as these are clearly in need of modernization.

A good case for further harmonization can also be made with respect to the option to impose a threshold on companies' right to identify shareholders. Member State interest is taking up this option has been quite limited and removing this option would lead to a cost saving for those firms operating in multiple markets that have had to adapt their systems to accommodate local thresholds. It would mean some erosion of shareholder privacy but modestly extending the rights of companies to identify their shareholders seems unlikely to entail significant corporate governance downsides.

A shareholder rights *regulation* rather than another directive would promote more consistent implementation and application on the ground. However, when what became SRD II was proposed in 2014, the Commission noted that 'only some basic principles' regarding shareholder identification, transmission of information and facilitation of the exercise of rights could be achieved because of the need to align with distinct corporate governance frameworks. National systems of corporate law and governance have not dramatically converged in the intervening years. This makes SRD III, perhaps as a 'maximum harmonization' directive coupled with another implementing act that aims to prevent diverging implementation at the detailed level the more likely form of a future reform package.

6. CONCLUSION

Shareholder engagement has come to be seen to be pivotal to good corporate governance. From that starting proposition it follows that it is more important than ever to consider whether the mechanisms of shareholder engagement are up to the task they are meant to perform. Historically, these mechanisms have fallen short. Custody chains perform valuable functions but they also create a distance between the company and the end investor from which flows a significant risk of voting preferences and other important information not passing smoothly up and down the chain. Corporate governance is thus at risk of being distorted by process deficiencies. The other side of the corporate governance coin is that custody chains can help to keep the identity of end investors hidden from companies. This can work to the advantage of activist investors but whether it is conducive to good corporate governance that serves the interests of society is debatable.

¹⁵⁰ ESMA (n 54), para 27.

¹⁵¹ ibid, paras 29-33.

¹⁵² AGC-EFC (n 63) advocates this.

¹⁵³ European Commission (n 118) 7.

The Shareholder Rights Directive II (SRD II) is an important intervention by the EU to tackle the corporate governance shortcomings of custody chains. One year on from these SRD II changes being transposed into national laws and becoming operative in practice (including in the UK, albeit only to a very limited extent), this article has found a mixed picture. The post-trade industry has taken the opportunity to upgrade shareholder identification, voting and related services offered to clients, the transition to fully electronic, latest market standards communications is underway, and there have been technological innovations. But experience has also shown that SRD II has not fully delivered on the necessary degree of harmonization, especially with respect to who is the 'shareholder' for this purpose. Shares are creations of national company and property laws and there are differences across EU/UK national legal regimes as to where control of shareholder rights sits within a custody chain. This has created problems for the application of SRD II in practice. SRD II has therefore struggled on two levels: it has created new uncertainties within national legal systems; and it has manifestly failed to achieve cross-border harmonization on a pivotal concept.

SRD II is based on improving custody chains rather than eliminating them. At a time of great excitement about the potential of new technologies, it is tempting to look to digital solutions that could eventually do away with custody chains and their associated problems for the operation of corporate governance. But custody chains are part of a complex post-trade ecosystem and they do provide significant benefits as well as some downsides. The digital transformation of an ecosystem – especially one whose very purpose is to take risk out of the financial system - is a mammoth undertaking. Technical complexities, uncertain business cases for ditching costly legacy systems and moving to costly new technologies, policymakers and regulators' intense focus on ensuring that the facilitation of innovation does not come at a cost of insufficient safeguards against risky practices that threaten public welfare, potentially huge changes to securities, company and property laws across multiple jurisdictions, and differing investors' preferences with respect to loss of anonymity are among the factors that will dictate the pace of the digital (r)evolution. A more gradual pace of change will likely involve new technologies being embraced first to improve the functioning of the existing system, including custody chains. Blockchain and API interfaces are already being offered by leading providers to help intermediaries meet their existing SRD II compliance obligations and there are also important public sector/central bank-led initiatives in this field.

Ruling out the possibility that profound technological change will eliminate custody chains in the short term makes addressing the shortcomings of SRD II an all the more pressing policy issue for the EU/EEA. The UK now has the regulatory autonomy to adopt solutions that are tailored for its domestic situation but those calling for reform of the law in the UK to better facilitate the participation of end investors in shareholder voting and other engagement activity, as well as internationally-active intermediaries for which idiosyncratic local practices represent operational burdens, can be expected to keep a close eye on the trajectory of EU law on these matters.